% Template for Electronic Journal of Graph Theory and Applications
% This file must be in the same folder as EJGTAart.sty and EJGTAhead.jpg

\documentclass[12pt]{elsarticle}
\usepackage[left=1in,right=1in, top=1.2in,bottom=1.2in]{geometry}
\usepackage{EJGTAart}
\usepackage{times}
\usepackage{amssymb,amsthm,latexsym,amsmath,epsfig,pgf}

% Set the volume if you know.
\volume{{\bf x} (x)}

% Set the starting page
\firstpage{1}

% Title (or short title) and author name for the header
\runauth{
On total edge product cordial labeling of fullerenes
\hspace{2ex} $\arrowvert$\hspace{2ex}
Martin Ba\v{c}a et al.}


% Put new theorem or any other settings for your document here
\newtheorem{theorem}{Theorem}[section]
\newtheorem*{theorem A}{Theorem A}
\newtheorem*{theorem B}{N\"olker's Theorem}
\newtheorem{lemma}{Lemma}[section]
\newtheorem{proposition}{Proposition}[section]
\newtheorem{corollary}{Corollary}[section]
\newtheorem{problem}{Problem}
\newtheorem*{question}{Question}
\newtheorem {conjecture}{Conjecture}
\theoremstyle{remark}
\newtheorem{remark}{Remark}[section]
\theoremstyle{remark}
\newtheorem{remarks}{Remarks}

\def\ml{l\kern-0.45ex\raise0.1ex\hbox{'}\kern-0.10ex}



\begin{document}
\begin{thebibliography}{99}


\bibitem{AHAL} A. Azaizeh, R. Hasni, A. Ahmad and G.C. Lau, $3$-total edge product cordial labeling of graphs, {\it Far East J. Math. Sci.} \textbf{96(2)} (2015), 193--209.

\bibitem{Cahit} I. Cahit, Cordial graphs: a weaker version of graceful and harmonious graphs, {\it Ars Combin.} \textbf{23} (1987), 201--207.

\bibitem{Cahit-37-90} I. Cahit, On cordial and $3$-equitable labelings of graphs, {\it  Utilitas Math.} {\bf 37} (1990), 189-–198.

\bibitem{Cairnie-216} N. Cairnie and K. Edwards, The computational complexity of cordial and equitable
labelling, {\it Discrete Math.} {\bf 216} (2000), 29--34.

\bibitem{Collins-90} K.L. Collins and M. Hovey, Most graphs are edge-cordial, {\it Ars Combin.} {\bf 30} (1990), 289-–295.

\bibitem{Deza-40-00} M. Deza, P.W. Fowler, A. Rassat and  K.M. Rogers, Fullerenes as tilings of surfaces, {\it J. Chem. Inf. Comput. Sci.}  {\bf 40} (2000), 550--558.

\bibitem{Ho-Lee-90} Y.S. Ho, S.M. Lee and S.C. Shee, Cordial labelings of the Cartesian product and composition of graphs, {\it Ars Combin.} {\bf 29} (1990), 169-–180.

\bibitem{Ho-Shee-93} Y.S. Ho and S.C. Shee, The cordiality of one-point union of $n$ copies of a graph, {\it Discrete Math.} {\bf 117} (1993), 225–-243.

\bibitem{Hovey-91} M. Hovey, A-cordial graphs, {\it Discrete Math.} {\bf 93} (1991),  183–-194.

\bibitem{Kirchherr-91} W.W. Kirchherr, On the cordiality of some specific graphs, {\it   Ars Combin.} {\bf 31} (1991), 127-–137.

\bibitem{Kirchherr-93} W.W. Kirchherr, NEPS operations on cordial graphs, {\it Discrete Math.} {\bf 115} (1993), 201-–209.

\bibitem{Klein-34-94} D.J. Klein, Elemental benzenoids, {\it J. Chem. Inf. Comput. Sci.}  {\bf 34} (1994), 453--459.

\bibitem{Kuoa-169-97} D. Kuo, G.J. Chang and Y.H.H. Kwong, Cordial labeling of $mK_n$, {\it Discrete Math.} {\bf 169} (1997), 121–-131.

\bibitem{Lee-Liu-91} S.M. Lee and A. Liu, A construction of cordial graphs from smaller cordial graphs, {\it Ars Combin.} {\bf 32} (1991), 209-–214.

\bibitem{SPS} M. Sundaram, R. Ponraj and S. Somasundaram, Product cordial labeling of graphs, {\it Bull.   Pure   Appl. Sci. Sect. E Math. Stat.} \textbf{23} (2004), 155--163.

\bibitem{Terrones-5-03} H. Terrones and M. Terrones, Curved nanostructured materials, {\it New J. Phys.} \textbf{5} (2003), 1--126.

\bibitem{VB} S.K. Vaidya and C.M. Barasara, Edge product cordial labeling of graphs, {\it J.  Math.  Comput. Sci.} \textbf{2(5)} (2012), 1436--1450.

\bibitem{VB1} S.K. Vaidya and C.M. Barasara, Total edge product cordial labeling of graphs, {\it Malaya J.   Matematik} \textbf{3(1)} (2013), 55--63.

\bibitem{West} D.B. West, Introduction to Graph Theory, 2nd Edition, Prentice-Hall, New Jersey, USA, (2003).
\end{thebibliography}


\begin{frontmatter}
% Write your paper title here
\papertitle{On total edge product cordial labeling of fu\-lle\-re\-nes}

%% use optional labels to link authors explicitly to addresses. The label may be more than one, use comma to separate

%\author[label1,label]{Edy Tri Baskoro${}^1$}

\author[label1]{Martin Ba\v{c}a}
\author[label2]{Muhammad Irfan}
\author[label2]{Aisha Javed}
\author[label1]{Andrea Semani\v{c}ov\'{a}-Fe\-\v{n}ov\-\v{c}\'{i}\-ko\-v\'a}


\address[label1]{\small Department of Applied Mathematics and Informatics, Technical University, Ko\v{s}ice, Slovak Republic}
\address[label2]{\small Abdus Salam School of Mathematical Sciences,   GC University, Lahore, Pakistan


\vspace*{2.5ex} 
 {\normalfont   martin.baca@tuke.sk, m.irfan.assms@gmail.com, aaishajaved@gmail.com, andrea.fenovcikova@tuke.sk}
 }

\begin{abstract}
For a simple graph $G=(V(G),E(G))$ this paper deals with the existence of an edge labeling  $\varphi :E(G) \to \{ 0,1,\dots,k - 1\} $, $2\leq k\leq |E(G)|$, which induces a vertex labeling ${\varphi ^*}:V(G) \to \{ 0,1,\dots,k - 1\} $ in such a way that for each vertex $v$, assigns the label $\varphi(e_1)\cdot\varphi(e_2)\cdot\ldots\cdot \varphi(e_n) \pmod k$, where $e_1, e_2,\dots, e_n$ are the edges incident to the vertex $v$. The labeling $\varphi$ is called a $k$-total edge product cordial labeling of $G$ if $|(e_{\varphi}(i)+v_{\varphi^*}(i))-(e_{\varphi}(j)+v_{\varphi^*}(j))|  \le 1$ for every $i,j$, $0 \le i < j \le k-1$, where $e_{\varphi}(i)$ and $v_{\varphi^*}(i)$ is the number of edges and vertices  with $\varphi(e)=i$ and $\varphi^*(v)=i$, respectively.

 The paper examines the existence of such labelings for toroidal fullerenes and for Klein-bottle fullerenes.

\let\thefootnote\relax\footnotetext{Received: xx xxxxx 20xx,\quad
  Accepted: xx xxxxx 20xx.\\[3ex]
  }
  
\end{abstract}

\begin{keyword}
% Separate keyword by \sep
cordial labeling \sep $k$-total edge product cordial labeling \sep toroidal fullerenes \sep  Klein-bottle fullerenes

% Write the classification number
Mathematics Subject Classification: 05C78 
\end{keyword}

\end{frontmatter}

%% Main text
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Introduction}
Let $G=(V(G),E(G))$ be a finite graph without loops and multiple edges, where $V(G)$ and $E(G)$ are its vertex set and edge set, respectively. General reference for graph-theoretic notions is \cite{West}.

A labeling of a graph is any mapping that sends some set of graph elements to a set of numbers or colors. If we label only vertices (respectively edges), we call such a labeling a {\it vertex} (respectively {\it edge}) {\it labeling}.

A vertex labeling $\phi: V(G)\to \{0,1\}$ induces an edge labeling $\phi^*: E(G)\to \{0,1\}$ defined by $\phi^*(uv)= |\phi(u)-\phi(v)|$. For a vertex labeling $\phi$ and $i\in \{0,1\}$, a vertex $v$ is an $i$-vertex if $\phi(v)=i$ and an edge $e$ is an $i$-edge if $\phi^*(e)=i$. Denote the numbers of $0$-vertices, $1$-vertices, $0$-edges, and $1$-edges of $G$ under $\phi$ and $\phi^*$ by $v_{\phi}(0)$, $v_{\phi}(1)$, $e_{\phi^*}(0)$, and $e_{\phi^*}(1)$, respectively. A vertex labeling $\phi$ is called {\it cordial} if $|v_{\phi}(0)-v_{\phi}(1)|\le 1$ and $|e_{\phi^*}(0)-e_{\phi^*}(1)|\le 1$. 

The notion of the cordial labeling was first introduced by Cahit \cite{Cahit} as a weaker version of graceful labeling. He proved in \cite{Cahit-37-90} that every tree is cordial, the complete bipartite graph $K_{m,n}$ is cordial for all $m$ and $n$, and the complete graph $K_n$ is cordial if and only if $n\le 3$. Cordial labelings of various families of graphs were studied in \cite{Ho-Lee-90,Kirchherr-91,Kuoa-169-97}. For related results see \cite{Ho-Shee-93,Lee-Liu-91} and for generalizations see  \cite{Collins-90,Hovey-91}. Cairnie and Edwards \cite{Cairnie-216} have determined the computational complexity of cordial labelings. They proved the conjecture of Kirchherr \cite{Kirchherr-93} that deciding whether a graph admitting a cordial labeling is NP-complete.

A binary vertex labeling $\phi: V(G)\to \{0,1\}$ with induced edge labeling $\phi^*: E(G)\to \{0,1\}$ defined by $\phi^*(uv)= \phi(u)\phi(v)$ is called a {\it product cordial labeling} if $|v_{\phi}(0)-v_{\phi}(1)|\le 1$ and $|e_{\phi^*}(0)-e_{\phi^*}(1)|\le 1$. The concept of the product cordial labeling was introduced by Sundaram et al. \cite{SPS}. Some labelings with variations in cordial theme, namely an edge product cordial labeling and  a total edge product cordial  labeling have been introduced by Vaidya and Barasara in \cite{VB,VB1}.

Let $k$ be an integer, $2 \le k \le |E(G)|$. An edge labeling $\varphi: E(G)\to \{0,1,\dots,k-1\}$ with induced vertex labeling $\varphi^*: V(G)\to \{0,1, \dots,k-1\}$ defined by $\varphi^*(v)= \varphi(e_1)\cdot\varphi(e_2)\cdot\ldots\cdot \varphi(e_n) \pmod k$, where $e_1, e_2,\dots, e_n$ are the edges incident to the vertex $v$, is called a {\it $k$-total edge product cordial labeling} of $G$ if $|(e_{\varphi}(i) + v_{\varphi^*}(i))- (e_{\varphi}(j) + v_{\varphi^*}(j))|\le 1$ for every $i,j$, $0 \le i < j \le k-1$. 

The concept of the $k$-total edge product cordial labeling was introduced by Azaizeh et al. in \cite{AHAL}. A~graph $G$ with a $k$-total edge product cordial labeling is called a {\it $k$-total edge product cordial graph}.

In the paper, we investigate the existence of $3$-total edge product cordial labeling for toroidal fullerenes and for Klein-bottle fullerenes. 

The discovery of the fullerene molecules and related forms of carbon such as nanotubes has ge\-ne\-ra\-ted an~explosion of activity in chemistry, physics, and materials science. Classical fullerene is an~all-carbon molecule in which the atoms are arranged on a~pseudospherical framework made up entirely of pentagons and hexagons. Its molecular graph is a~finite trivalent graph embedded on the surface of a~sphere with only hexagonal and (exactly $12$) pentagonal faces. Deza et al. \cite{Deza-40-00} considered fullerene's extension to other closed surfaces and showed that only four surfaces are possible, namely sphere, torus, Klein bottle and projective plane. Unlike spherical fullerenes, toroidal and Klein bottle's fullerenes have been regarded as tessellations of entire hexagons on their surfaces since they must contain no pentagons, see \cite{Deza-40-00, Klein-34-94}.

Let $L$ be a~regular hexagonal lattice and let $P_m^n$ be an~$m\times n$ quadrilateral section (with $m\ge 2$ hexagons on the top and bottom sides and $n\ge 2$ hexagons on the lateral sides, $n$ is even) cut from the regular hexagonal lattice $L$, (see Figure~\ref{obr1}). 

\begin{figure}[ht!]
	\centering
		\includegraphics[scale=.7]{figure2/figure101.pdf}
	\caption{Quadrilateral section $P_m^n$ cut from the regular  hexagonal lattice $L$.}
	\label{obr1}
\end{figure}

If we identify two lateral sides of $P_m^n$ then we form a~cylinder. If we identify the top and bottom sides of the cylinder such that we identify the vertices $u_i^0$ and $u_i^{n}$, and the vertices $v_i^0$ and $v_i^{n}$, for $i=1,2,\dots, m$, we are able to obtain the {\it toroidal fullerene} (toroidal polyhex) $\mathbb{H}_m^n$ with $mn$ hexagons. We can see that the  toroidal fullerene is a~cubic bipartite graph embedded on the torus such that each face is a~hexagon. If we identify the top and bottom sides of the cylinder in such a~way that we identify the vertices $u_1^0$ and $u_1^{n}$, the vertices $u_i^0$ and $u_{m+2-i}^{n}$, for $i=2,3,\dots, m$, and the vertices $v_i^0$ and $v_{m+1-i}^{n}$, for $i=1,2,\dots, m$, we  obtain the {\it Klein-bottle fullerene} (Klein-bottle polyhex) $\mathbb{KB}_m^n$ with $mn$ hexagons. In this case $\mathbb{KB}_m^n$ is a~cubic bipartite graph of order $2mn$ and size $3mn$ embedded on the Klein-bottle and contains only hexagons.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Product cordial labeling of toroidal polyhex $\mathbb{H}_m^n$}

In the following segments  by an {\it open edge} we mean the edge with only one end vertex. We will use two operations for gluing the segments, where the symbol $\oplus_v$ denotes the operation of gluing two segments  in the vertical direction and analogously  the symbol $\oplus_h$ denotes the operation of gluing two segments in the horizontal direction. 
 
 The next theorem shows that the toroidal polyhex $\mathbb{H}_3^n$,   $n$ even, admits a $3$-total edge product cordial labeling.

\begin{theorem}    \label{theorem for H-3-n}
 For $n$ even, $n\ge 4$, the toroidal polyhex $\mathbb{H}_3^n$ is $3$-total edge product cordial.
\end{theorem}
\begin{proof}
For obtaining the toroidal polyhex $\mathbb{H}_3^n$ we will use the labeled segments $A_3^2$, $B_3^2$ and $C_3^2$ illustrated in Figure \ref{segment A-3-2, B-3-2, C-3-2}. We can see that each labeled segment has the same number of 0, 1 and 2, namely in the segment $A_3^2$ every number of zeros, ones and twos is used 14 times, in the segment $B_3^2$ every number is used 10 times and in the segment $C_3^2$ every number is used 6 times as a label. 

   \begin{figure}[ht!]
 \centering \includegraphics{figure2/figure1}  \hspace{1.08cm} \includegraphics{figure2/figure2}
 
 \bigskip
 \centering \includegraphics{figure2/figure3}
  \caption{The labeled segments $A_3^2$, $B_3^2$ and $C_3^2$.}
 \label{segment A-3-2, B-3-2, C-3-2}
  \end{figure}

%   \begin{figure}[ht!]
% \centering \includegraphics{figure-A-3-2}
%  \caption{The labeled segments $A_3^2$, $B_3^2$ and $C_3^2$.}
% \label{segment A-3-2, B-3-2, C-3-2}
%  \end{figure}

First we glue $(\frac{n}{2}-2)$  segments $B_3^2$ together in the vertical direction. Since the open edges in the segment $B_3^2$ are labeled with number 1 it follows that by gluing these segments we do not change the vertex labels in the segment   $B_3^2 \oplus_v B_3^2 \oplus_v \dots \oplus_v B_3^2 = (\frac{n}{2}-2)B_3^2$. Then we glue in the vertical direction the segment $A_3^2$ to the segment $(\frac{n}{2}-2)B_3^2$ to obtain  $A_3^2 \oplus_v (\frac{n}{2}-2)B_3^2$. Finally we glue the segment $A_3^2 \oplus_v (\frac{n}{2}-2)B_3^2$ in the vertical direction to the segment $C_3^2$. All open edges used in the gluing  operations are labeled with the number 1 therefore these operations do not have any impact to the vertex labels in the resulting segment $H_3^n$, where
$$H_3^n =    \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                         (\frac{n}{2}-2)B_3^2 \\
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix}.$$

Now we identify two lateral sides of the segment $H_3^n$ to form a~cylinder or nanotube.  Then we identify the top and bottom sides of the cylinder such that we join the open edges in the bottom side of the cylinder labeled by $0$ to the corresponding vertices in the top side of the cylinder (labeled by $0$) and we obtain the toroidal polyhex $\mathbb{H}_3^n$ with $3n$ hexagons. Table \ref{table1} shows multiplicity of numbers $0$, $1$ and $2$ used in the segments $A_3^2$, $B_3^2$, $C_3^2$ and $H_3^n$. 

%Table 1

\begin{table}[ht!]
	\renewcommand{\arraystretch}{1.5}
\setlength{\tabcolsep}{2.5pt}
\centering
		\begin{tabular}{|l||l|l|l|}
		\hline
		segment  & $e_{\varphi}(0)+v_{\varphi^*}(0)$ &  $e_{\varphi}(1)+v_{\varphi^*}(1)$& $e_{\varphi}(2)+v_{\varphi^*}(2)$\\
		\hline\hline
		$A_3^2$& $14$&$14$&$14$\\
		\hline
		$B_3^2$& $10$&$10$&$10$\\
		\hline
		$C_3^2$& $6$&$6$&$6$\\
		\hline
		$(\frac{n}{2}-2)B_3^2$& $5n-20$&$5n-20$&$5n-20$\\
		\hline
		$H_3^n = A_3^2 \oplus_v (\frac{n}{2}-2)B_3^2 \oplus_v C_3^2$& $5n$ & $5n$& $5n$\\
		\hline
			\end{tabular}
	\caption{Multiplicity of     0s,   of 1s and   of 2s in the segments 	$A_3^2$, 	$B_3^2$, 	$C_3^2$, 	$(\frac{n}{2}-2)B_3^2$ and $H_3^n$.}
	\label{table1}
\end{table}		

It is only routine checking that the toroidal polyhex $\mathbb{H}_3^n$ contains every number of 0, 1 and 2 exactly $5n$ times.
\end{proof}

Next we extend the construction of the segment $H_3^n$ for arbitrary $m$ to obtain the segment $H_m^n$ and then to obtain the  graph of the toroidal polyhex $\mathbb{H}_m^n$.


\begin{theorem}    \label{theorem for H-m-n}
 For $n$ even, $n\ge 4$ and $m\ge 3$ the, toroidal polyhex $\mathbb{H}_m^n$ is $3$-total edge product cordial.
\end{theorem}
\begin{proof}
For obtaining a $3$-total edge product cordial labeling of $\mathbb{H}_m^4$ we need the next labeled segments $D_1^4$ and $D_2^4$ depicted in Figure \ref{segment D-1-4, D-2-4}.

   \begin{figure}[ht!]
 \centering \includegraphics{figure2/figure5}  \hspace{1.08cm} \includegraphics{figure2/figure12}
  \caption{The labeled segments $D_1^4$ and $D_2^4$.}
 \label{segment D-1-4, D-2-4}
  \end{figure}

For $m=3t$, $t\ge 1$, we glue the segment $A_3^2$ in the vertical direction to the segment $C_3^2$ and the resulting segment $A_3^2 \oplus_v C_3^2$ we glue in the horizontal direction $t$ times. Hence 
$$H_m^4 =   {\underbrace{ \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix} 
		                      \oplus_h
		                       \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix} 
		                      \oplus_h
		                      \cdots
		                      \oplus_h
		                       \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix}}_{t} }.
		                      $$
For $m=3t+1$, $t\ge 1$, we glue the segment $  \left[ A_3^2 \oplus_v C_3^2 \right]$ horizontally $t$ times and moreover, we glue horizontally the segment $D_1^4$ and we obtain  $$H_m^4 = {\underbrace{ \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix} 
		                      \oplus_h
		                       \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix} 
		                      \oplus_h
		                      \cdots
		                      \oplus_h
		                       \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix}}_{t} }\oplus_h D_1^4.$$
For $m=3t+2$, $t\ge 1$, we glue the segment $  \left[ A_3^2 \oplus_v C_3^2 \right]$  horizontally $t$ times and moreover, we glue horizontally  the segment $D_2^4$ and we create the segment  $$H_m^4 = {\underbrace{ \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix} 
		                      \oplus_h
		                       \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix} 
		                      \oplus_h
		                      \cdots
		                      \oplus_h
		                       \begin{bmatrix} 
		                     {A}_3^2\\ 
		                         \oplus_v\\ 
		                               C_3^2
		                      \end{bmatrix}}_{t} }\oplus_h D_2^4.$$

Table \ref{table2} shows how many times the numbers $0$, $1$ and $2$ are used as edge and vertex labels in the segments $D_1^4$ and $D_2^4$ and in the resulting segment  $H_m^4$.

%Table 2

\begin{table}[ht!]
	\renewcommand{\arraystretch}{1.5}
\setlength{\tabcolsep}{2.5pt}
\centering
		\begin{tabular}{|l||l|l|l|}
		\hline
		segment  & $e_{\varphi}(0)+v_{\varphi^*}(0)$ &  $e_{\varphi}(1)+v_{\varphi^*}(1)$& $e_{\varphi}(2)+v_{\varphi^*}(2)$\\
		\hline\hline
		$D_1^4$& $6$&$7$&$7$\\
		\hline
		$D_2^4$& $14$&$13$&$13$\\
		\hline
		$H_m^4$, $m=3t$, $t\ge 1$& $20t$&$20t$&$20t$\\
		\hline
			$H_m^4$, $m=3t+1$, $t\ge 1$& $20t+6$&$20t+7$&$20t+7$\\
		\hline
			$H_m^4$, $m=3t+2$, $t\ge 1$& $20t+14$&$20t+13$&$20t+13$\\
		\hline
			\end{tabular}
	\caption{Multiplicity of     0s,   of 1s and   of 2s in the segments 	$D_1^4$, 	$D_2^4$ and $H_m^4$.}
	\label{table2}
\end{table}		

One can see that the resulting segments $H_m^4$ in each previous case satisfies the property of having a  $3$-total edge product cordial labeling. We identify two lateral sides of the segment $H_m^4$ to form a~cylinder and then we identify the top and bottom sides of the cylinder such that we join the open edges in the bottom side of the cylinder to the corresponding vertices in the top side of the cylinder labeled by $0$. Since the open edges in the corresponding segments are  labeled with number 1 (except for open edges incident to vertices in the top side) it follows that gluing these segments does not have any impact on the vertex labels in the resulting toroidal polyhex $\mathbb{H}_m^4$.  

The graph $\mathbb{H}_m^n$ with a $3$-total edge product cordial labeling for $n$ even, $n\ge 6$, we obtain by using the segment $H_3^n = A_3^2 \oplus_v (\frac{n}{2}-2)B_3^2 \oplus_v C_3^2$ from Theorem \ref{theorem for H-3-n} such that the certain multiple of the segment $H_3^n$ we glue in horizontal direction with a special segment   which depends on   $n$ and $m$.

For next operations we need new segments $A_1^4$, $C_2^4$,  $C_1^2$ and $C_1^4$ depicted in Figure \ref{segment c-1-2, C-1-4} and the segments $B_1^6$, $C_1^6$ and $C_2^6$ shown in Figure \ref{segment B-1-6, C-1-6, C-2-6}.


   \begin{figure}[ht!]
 \centering \includegraphics{figure2/figure4}  \hspace{1.08cm} \includegraphics{figure2/figure11}
 % \caption{The labeled segments $A_1^4$ and $C_2^4$.}
% \label{segment A-1-4, C-2-4}
%  \end{figure}
%
%   \begin{figure}[ht!]
 \hspace{1.08cm} \includegraphics{figure2/figure6}  \hspace{1.08cm} \includegraphics{figure2/figure7}
  \caption{The labeled segments  $A_1^4$, $C_2^4$,   $C_1^2$ and $C_1^4$.}
 \label{segment c-1-2, C-1-4}
  \end{figure}

   \begin{figure}[ht!]
 \centering \includegraphics{figure2/figure8}  \hspace{1.08cm} \includegraphics{figure2/figure9}     \hspace{1.08cm} 
 \includegraphics{figure2/figure10}
  \caption{The labeled segments  $B_1^6$, $C_1^6$ and $C_2^6$.}
 \label{segment B-1-6, C-1-6, C-2-6}
  \end{figure}

Table \ref{table3} gives multiplicity of numbers 0, 1 and 2 used in the segments $A_1^4$, $C_2^4$, $C_1^2$, $C_1^4$, $B_1^6$, $C_1^6$ and $C_2^6$.

%Table 3

\begin{table}[ht!]
	\renewcommand{\arraystretch}{1.5}
\setlength{\tabcolsep}{2.5pt}
\centering
		\begin{tabular}{|l||l|l|l|}
		\hline
		segment  & $e_{\varphi}(0)+v_{\varphi^*}(0)$ &  $e_{\varphi}(1)+v_{\varphi^*}(1)$& $e_{\varphi}(2)+v_{\varphi^*}(2)$\\
		\hline\hline
		$A_1^4$& $8$&$8$&$8$\\
		\hline
		$C_2^4$& $10$&$11$&$11$\\
		\hline
		$C_1^2$& $2$&$2$&$2$\\
		\hline
		$C_1^4$& $5$&$6$&$5$\\
		\hline
		$B_1^6$& $10$&$10$&$10$\\
		\hline
		$C_1^6$& $9$&$9$&$8$\\
		\hline
		$C_2^6$& $17$&$18$&$17$\\
		\hline
			\end{tabular}
	\caption{Multiplicity of     0s,   of 1s and   of 2s in the segments 	$A_1^4$, $C_2^4$, $C_1^2$, $C_1^4$, $B_1^6$, $C_1^6$ and $C_2^6$.}
	\label{table3}
\end{table}		

Let us consider the following 7 cases.

{\it Case 1.} For $m=3t$, $t\ge 1$, and every $n$ even, $n\ge 6$, we get
$$ H_m^n =  \underbrace{H_3^n \oplus_h H_3^n \oplus_h \cdots \oplus_h H_3^n}_{t}     .$$

{\it Case 2.} For $m=3t+1$, $t\ge 1$, and $n=6s$, $s\ge 1$, the special segment   we obtain by gluing vertically the segment $A_1^4$ with vertically $(s-1)$ times of the segment $B_1^6$ and the resulting segment $A_1^4 \oplus_v (s-1)B_1^6$ we glue in the vertical  direction with the segment $C_1^2$. Then we have
$$ H_m^n =  \left[ \underbrace{H_3^n \oplus_h H_3^n \oplus_h \cdots \oplus_h H_3^n}_{t}\right] \oplus_h \begin{bmatrix} 
		                     A_1^4\\ 
		                         \oplus_v\\ 
		                              (s-1)B_1^6\\ 
		                         \oplus_v\\  
		                         C_1^2
		                      \end{bmatrix} .$$

{\it Case 3.} For $m=3t+2$, $t\ge 1$, and $n=6s$, $s\ge 1$, the segment $H_m^n$ we obtain as follows 
$$ H_m^n =  \left[  \underbrace{H_3^n \oplus_h H_3^n \oplus_h \cdots \oplus_h H_3^n}_{t} \right] \oplus_h   
\begin{bmatrix} 
		                     A_1^4\\ 
		                         \oplus_v\\ 
		                              (s-1)B_1^6\\ 
		                         \oplus_v\\  
		                         C_1^2
		                      \end{bmatrix}\oplus_h   
\begin{bmatrix} 
		                     A_1^4\\ 
		                         \oplus_v\\ 
		                              (s-1)B_1^6\\ 
		                         \oplus_v\\  
		                         C_1^2
		                      \end{bmatrix} .$$

{\it Case 4.} For $m=3t+1$, $t\ge 1$, and $n=6s+2$, $s\ge 1$, the special segment   we create by gluing the segment $A_1^4$ in the vertical  direction with vertically  $(s-1)$ times of the segment $B_1^6$ and the resulting segment $A_1^4 \oplus_v (s-1)B_1^6$ we glue in the  vertical  direction with the segment $C_1^4$. Then we get
$$ H_m^n =  \left[  \underbrace{H_3^n \oplus_h H_3^n \oplus_h \cdots \oplus_h H_3^n}_{t}\right] \oplus_h \begin{bmatrix} 
		                     A_1^4\\ 
		                         \oplus_v\\ 
		                              (s-1)B_1^6\\ 
		                         \oplus_v\\  
		                         C_1^4
		                      \end{bmatrix} .$$

{\it Case 5.} For $m=3t+2$, $t\ge 1$, and $n=6s+2$, $s\ge 1$, the special segment  we obtain by gluing two segments $A_1^4$ in the vertical  direction with vertically $2(s-1)$ times of the segment $B_1^6$ and the resulting segment $2A_1^4 \oplus_v 2(s-1)B_1^6$ we glue in the  vertical  direction with the segment $C_2^4$. Then we have
$$ H_m^n =\left[  \underbrace{H_3^n \oplus_h H_3^n \oplus_h \cdots \oplus_h H_3^n}_{t}\right] \oplus_h \begin{bmatrix} 
		                    2 A_1^4\\ 
		                         \oplus_v\\ 
		                             2 (s-1)B_1^6\\ 
		                         \oplus_v\\  
		                         C_2^4
		                      \end{bmatrix} .$$

{\it Case 6.} For $m=3t+1$, $t\ge 1$, and $n=6s+4$, $s\ge 1$, the special segment   we generate by gluing the segment $A_1^4$ in the vertical  direction with vertically $(s-1)$ times of the segment $B_1^6$ and the resulting segment $A_1^4 \oplus_v (s-1)B_1^6$ we glue in the  vertical  direction with the segment $C_1^6$. The expected resulting segment will be as follows 
$$ H_m^n = \left[  \underbrace{H_3^n \oplus_h H_3^n \oplus_h \cdots \oplus_h H_3^n}_{t}\right] \oplus_h \begin{bmatrix} 
		                     A_1^4\\ 
		                         \oplus_v\\ 
		                              (s-1)B_1^6\\ 
		                         \oplus_v\\  
		                         C_1^6
		                      \end{bmatrix} .$$

{\it Case 7.} For $m=3t+2$, $t\ge 1$, and $n=6s+4$, $s\ge 1$, the special segment   we create by gluing two segments $A_1^4$ in the vertical  direction with vertically $2(s-1)$ times of the segment $B_1^6$ and the resulting segment $2A_1^4 \oplus_v 2(s-1)B_1^6$ we glue in the  vertical  direction with the segment $C_2^6$. The required segment has the following form 
$$ H_m^n = \left[  \underbrace{H_3^n \oplus_h H_3^n \oplus_h \cdots \oplus_h H_3^n}_{t}\right] \oplus_h \begin{bmatrix} 
		                   2  A_1^4\\ 
		                         \oplus_v\\ 
		                            2  (s-1)B_1^6\\ 
		                         \oplus_v\\  
		                         C_2^6
		                      \end{bmatrix} .$$

%Table 4


\begin{table}[ht!]
	\renewcommand{\arraystretch}{1.5}
\setlength{\tabcolsep}{2.5pt}
\centering
		\begin{tabular}{|l|l||l|l|l|}
		\hline
	Case&	segment $H_m^n$ & $e_{\varphi}(0)+v_{\varphi^*}(0)$ &  $e_{\varphi}(1)+v_{\varphi^*}(1)$& $e_{\varphi}(2)+v_{\varphi^*}(2)$\\
		\hline\hline
	1.&	$m=3t$, $t\ge 1$& $5nt$&$5nt$&$5nt$\\
	&$n\ge 6$, $n$ even &&&\\
		\hline		
	2.&	$m=3t+1$, $t\ge 1$& $5nt+10s$&$5nt+10s$&$5nt+10s$\\
	&$n=6s$, $s\ge 1$ &&&\\
		\hline
	3.&	$m=3t+2$, $t\ge 1$& $5nt+20s$&$5nt+20s$&$5nt+20s$\\
	&$n=6s$, $s\ge 1$ &&&\\
		\hline
	4.&	$m=3t+1$, $t\ge 1$& $5nt+10s+3$&$5nt+10s+4$&$5nt+10s+3$\\
	&$n=6s+2$, $s\ge 1$ &&&\\
		\hline
	5.&	$m=3t+2$, $t\ge 1$& $5nt+20s+6$&$5nt+20s+7$&$5nt+20s+7$\\
	&$n=6s+2$, $s\ge 1$ &&&\\
		\hline
	6.&	$m=3t+1$, $t\ge 1$& $5nt+10s+7$&$5nt+10s+7$&$5nt+10s+6$\\
	&$n=6s+4$, $s\ge 1$ &&&\\
		\hline
	7.&	$m=3t+2$, $t\ge 1$& $5nt+20s+13$&$5nt+20s+14$&$5nt+20s+13$\\
	&$n=6s+4$, $s\ge 1$ &&&\\
		\hline
			\end{tabular}
	\caption{Multiplicity of     0s,   of 1s and   of 2s in the segment  	$H_m^n$ for $n$ even, $n\ge 6$ and $m\ge 3$.}
	\label{table4}
\end{table}		


All possible cases for obtaining the segment $H_m^n$ for $n$ even, $n\ge 6$ and $m\ge 3$, are described in Table \ref{table4}, where it is shown how many times the numbers 0, 1  and 2 are used as edge and vertex labels. 

Identifying two lateral sides of the segment $H_m^n$ we form a~cylinder and then identifying the top and bottom sides of the cylinder such that the open edges in the bottom side of the cylinder are joined to the corresponding vertices in the top side of the cylinder, we obtain the toroidal polyhex $\mathbb{H}_m^n$. We can see that the labels of the open edges used to glue the corresponding segments has no effect on the vertex labels in the resulting segments/graphs. From Table 4 it follows that for every case the resulting toroidal polyhex $\mathbb{H}_m^n$ is $3$-total edge product cordial.

\end{proof}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

A {\it helical torus} decorated with graphite can be formed by joining the opposite ends of a chiral or helical nanotube, such that the cycle of one end of nanotube is rotated relative to the cycle of the second end of nanotube, see \cite{Terrones-5-03}. More precisely, we suppose that a nanotube is our cylinder created identifying two lateral sides of $P_m^n$ in Figure \ref{obr1}. To create the helical torus we identify the top and bottom sides of the cylinder/nanotube such that we identify the vertices $u_i^0$ and $u_{i+j}^n$ and the vertices $v_i^0$ and $v_{i+j}^n$, for $i=1,2,\dots, m$ and $j=1,2, \dots, m-1$, where $i+j$ is taken modulo $m$. 

Since in the proof of Theorem \ref{theorem for H-m-n} the segment $H_m^n$, for $n$ even, $n\ge 4$ and $m\ge 3$, contains in the top side only the vertices labeled by $0$, it follows that any gluing of the open edges in the bottom side of $H_m^n$ to the vertices in the top side is possible and does not have any impact on the vertex labels in the resulting graph of the helical torus $\mathbb{H}_m^n$. It means that the gluing operation satisfies the property of having a $3$-total edge product cordial labeling. Hence we have 

\begin{theorem}    \label{theorem helical torus}
 For $n$ even, $n\ge 4$ and $m\ge 3$, the helical torus $\mathbb{H}_m^n$ admits a $3$-total edge product cordial labeling.
\end{theorem}

By joining the opposite ends of a nanotube we can obtain the {\it Klein-bottle polyhex}. More precisely, to create the Klein-bottle polyhex $\mathbb{KB}_m^n$ we identify the top and bottom sides of the cylinder/nanotube (see Figure \ref{obr1}) such that we identify the vertices  $u_1^0$ and $u_1^{n}$, the vertices $u_i^0$ and $u_{m+2-i}^{n}$, for $i=2,3,\dots, m$, and the vertices $v_i^0$ and $v_{m+1-i}^{n}$, for $i=1,2,\dots, m$. 

With respect to the fact that the gluing operation does not have any impact on the vertex labels in the resulting graph of the Klein-bottle polyhex $\mathbb{KB}_m^n$, therefore the property of admitting a $3$-total edge product cordial labeling also holds. Consequently we get 

\begin{theorem}    \label{theorem Klein-bottle}
 For $n$ even, $n\ge 4$ and $m\ge 3$, the Klein-bottle polyhex $\mathbb{KB}_m^n$ admits a $3$-total edge product cordial labeling.
\end{theorem}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Conclusion}

In this paper we proved the existence of the $3$-total edge product cordial labeling for the toroidal polyhex, respectively helical torus, $\mathbb{H}_m^n$, and for the Klein-bottle polyhex $\mathbb{KB}_m^n$, for $n$ even, $n\ge 4$ and $m\ge 3$. In this case the Klein-bottle polyhex $\mathbb{KB}_m^n$ is a~cubic bipartite graph. The next construction describes the existence of the  Klein-bottle polyhex $\mathbb{KB}_{m+1/2}^n$ as a~cubic non-bipartite graph.

Let $L$ be a~regular hexagonal lattice and  let $P_{m+1/2}^n$ be a~quadrilateral section (with $m+1/2$ hexagons on the top and bottom sides, $m\ge 1$, and $n\ge 2$ hexagons on the lateral sides, $n$ is even) cut from the regular hexagonal lattice $L$, (see Figure~\ref{obr2}). 

\begin{figure}[ht!]
	\centering
		\includegraphics[scale=.7]{figure2/figure102.pdf}
	\caption{Quadrilateral section $P_{m+1/2}^n$ cut  from the regular  hexagonal lattice $L$.}
	\label{obr2}
\end{figure}

We identify the top and bottom sides of $P_{m+1/2}^n$ to form a~cylinder. Then we identify the lateral sides of the cylinder such that we identify the vertices $u_1^0$ and $v_{m+1}^{0}$, and the vertices $u_1^j$ and $v_{m+1}^{n-j}$, for $j=1,2,\dots, n-1$, to obtain the  Klein-bottle polyhex   $\mathbb{KB}_{m+1/2}^n$. We can see that $\mathbb{KB}_{m+1/2}^n$ is a~cubic non-bipartite graph of order $2n(m+1/2)$ and size $3n(m+1/2)$ embedded on the Klein-bottle and contains $n(m+1/2)$ hexagons.

Let us suggest the following open problem.

\begin{problem} Find a   $3$-total edge product cordial labeling for the Klein-bottle polyhex  $\mathbb{KB}_{m+1/2}^n$, for $n$ even, $n\ge 2$ and $m\ge 3$.
\end{problem}



%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%
\section*{Acknowledgement}
This work was supported by the Slovak Science and Technology Assistance Agency under the contract No. APVV-15-0116.


%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%

\section*{References} 


\end{document}



\bibitem{Ahmad-AMC-243-14} A. Ahmad,   O.B.S. Al-Mushayt and M. Ba\v{c}a, {On Edge Irregularity Strength of Graphs}, {\it Appl. Math. Comput.} {\bf 243} (2014),   607--610.

\bibitem{Ahmad-Ars-114-14}  A. Ahmad and   M. Ba\v{c}a, {Total Edge Irregularity Strength of a~Categorical Product of Two Paths}, {\it Ars Combin.} {\bf 114} (2014),   203--212.

\bibitem{Ahmad-TCS-54-14}  A. Ahmad,  M. Ba\v{c}a and  M.K. Siddiqui, {On Edge Irregular Total Labeling of Categorical Product of Two Cycles}, {\it Theory Comp. Systems} {\bf 54}(1) (2014),  1--12.

\bibitem{Aigner-3-90} M. Aigner and   E. Triesch, {Irregular Assignments of Trees and Forests},  {\it SIAM J. Discrete Math.} {\bf 3} (1990), 439--449.

\bibitem{Amar-190-98} D. Amar and  O. Togni, {Irregularity Strength of Trees},  {\it Discrete Math.} {\bf 190} (1998), 15–-38.

\bibitem{Anholcer-309-09} M. Anholcer,    M. Kalkowski and J. Przyby\l{}o, {A New Upper Bound for the Total Vertex Irregularity Strength of Graphs}, {\it Discrete Math.} {\bf 309} (2009),  6316--6317.

\bibitem{Ashraf-8-4-2016}   F. Ashraf,   M. Ba\v{c}a, Z. Kim\'akov\'a and  A. Semani\v{c}ov\'{a}-Fe\v{n}ov\v{c}\'{i}kov\'a, {On Vertex and Edge $H$-irregularity Strengths of Graphs},  {\it Discrete Math. Algorithms and Applications} {\bf 8}(4) (2016), 13 pages.

\bibitem{ABLSF-Discuss}  F.  Ashraf,   M. Ba\v{c}a, M. Lascs\'akov\'a and  A. Semani\v{c}ov\'{a}-Fe\v{n}ov\v{c}\'{i}kov\'a, {On $H$-irregularity Strength of Graphs},  {\it Discuss. Math. Graph Theory}, accepted.

\bibitem{Baca-307-07} M.  Ba\v{c}a,   S. Jendro\ml, M. Miller and  J. Ryan, {On Irregular Total Labellings}, {\it Discrete Math.} {\bf 307} (2007),  1378--1388.

\bibitem{BaKa-235-14} M. Ba\v{c}a and   M.K. Siddiqui, {Total Edge Irregularity Strength of Generalized Prism}, {\it Applied Math. Comput.} {\bf 235} (2014), 168--173.

\bibitem{Brandt-57-08} S. Brandt,    J. Mi\v{s}kuf and  D. Rautenbach, {On a~Conjecture about Edge Irregular Total Labellings}, {\it J. Graph Theory} {\bf 57} (2008),  333--343.

\bibitem{Chartrand-64-88} G. Chartrand, M.S. Jacobson, J. Lehel, O.R. Oellermann, S. Ruiz and  F. Saba, Irregular Networks, {\it Congr. Numer.} {\bf 64} (1988),  187--192.

\bibitem{Haque-50-12} K.M.M. Haque,   {Irregular Total Labellings of Generalized Petersen Graphs}, {\it Theory Comp. Systems} {\bf 50} (2012),  537--544.

\bibitem{Ivanco-26-06} J. Ivan\v{c}o  and S.   Jendro\ml, {Total Edge Irregularity Strength of Trees}, {\it Discussiones Math. Graph Theory}  {\bf 26} (2006),  449--456.

\bibitem{Jendrol-28-07} S. Jendro\ml, J. Mi\v{s}kuf  and  R. Sot\'{a}k, Total Edge Irregularity Strength of Complete Graphs and Complete Bipartite Graphs, {\it Electron. Notes Discrete Math.}  {\bf 28} (2007), 281--285.

\bibitem{Jendrol-310-10} S. Jendro\ml, J. Mi\v{s}kuf  and  R. Sot\'{a}k, Total Edge Irregularity Strength of Complete Graphs and Complete Bipartite Graphs, {\it  Discrete Math.} {\bf 310} (2010), 400--407.

\bibitem{Kalkowski-25-11} M. Kalkowski, M. Karonski  and  F. Pfender, {A New Upper Bound for the Irregularity Strength of Graphs},  {\it SIAM J. Discrete Math.} {\bf 25} (2011),
1319-–1321.


\bibitem{Kinch-94-91} L. Kinch and  J. Lehel, {The Irregularity Strength of $tP_3$},  {\it Discrete Math.} {\bf 94} (1991), 75–-79.

\bibitem{Majerski-28-14} P. Majerski and  J. Przyby\l{}o, {On the Irregularity Strength of Dense Graphs}, {\it SIAM J. Discrete Math.} {\bf 28}(1) (2014), 197--205.

\bibitem{Majerski-76-14} P. Majerski  and J. Przyby\l{}o, {Total Vertex Irregularity Strength of Dense Graphs}, {\it J. Graph Theory} {\bf 76}(1) (2014), 34--41.

\bibitem{MSM} C.C. Marzuki, A.N.M. Salman and M. Miller, On the Total Irregularity Strength on Cycles and Paths, {\it Far East J. Math. Sci.} {\bf 82}(1)  (2013), 1–-21.


\bibitem{Nurdin-65-08} Nurdin, A.N.M. Salman and E.T. Baskoro, {The Total Edge-irregular Strengths of the Corona Product of Paths with some Graphs}, {\it J. Combin. Math. Combin. Comput.} {\bf 65} (2008),  163--175.

\bibitem{Przybylo-23-09} J.  Przyby\l{}o,   {Linear Bound on the Irregularity Strength and the Total Vertex Irregularity Strength of Graphs}, {\it SIAM J. Discrete Math.} {\bf 23} (2009),  511--516.


\bibitem{Ramdani-AKCE-10-13}  R. Ramdani and A.N.M. Salman, {On the Total Irregularity Strength of Some Cartesian Product Graphs}, {\it AKCE Int. J. Graphs Comb.} {\bf 10}(2) (2013),  199--209.

\end{thebibliography}

\end{document}

%%
%% End of file `ecrc-template.tex'. 